C

S

3

1 8 : O

p

e r a t i n g Sy

ste ms

I n s t r u ct or

R ic h ar d

J

. Li pton

G e

ne r

a l In f

or ma

t

ion Th e text

book

for this term is\

ModernOperating

S

ystems " by Tanenbaum.Wewi ll be going over the�rst six chapters in detail. We will survey some of

the other operating systems at the end of the book. We assume that you are

comfortable withC.

Generally

, we will go over and add to the material from the textbook.Problem

sets will be handed out onFridays and will be due in one or two weeks depending

on their complexity.Your grades will be based as follows:

(These weights are

subject to minor changes). Problem sets:4

0 %

Mid Term:20%

Final: 30%

Other: 10%

F i r s t Assi g

n m ent

: Please begin reading chapter two of the textbook on

\processes". He l

p: Students can discuss lectures and problem sets with each other but work

turned in must represent the individual student’s work. If you have anyques

-

tions, feel free to send e-mail to us:

Richard Lipton rjl@princeton.edu

Sanjeev

Kumar kumar@princeton.edu

The secretary for the course is Melissa Lawson. You can get her via e-mail at

mml@princeton.edu.

Her o�ce is on the 3rd oor. She will be holding her o�ce

hours at TB

A. Lipton’s o�ce hours will also posted shortly. 1